Author's response to reviews

Title: Revising acute care systems and processes to improve breastfeeding and maternal postnatal health: A pre and post intervention study in one English maternity unit

Authors:

Debra E Bick (debra.bick@kcl.ac.uk)
Trevor Murrells (trevor.murrells@kcl.ac.uk)
Annette Weavers (annette.weavers@royalberkshire.ac.uk)
Val Rose (val.rose@royalberkshire.ac.uk)
Julie Wray (j.wray@salford.ac.uk)
Sarah Beake (sarah.beake@kcl.ac.uk)

Version: 3 Date: 16 April 2012

Author's response to reviews:

Dear Sir/Madam

We are pleased to submit our revised paper and thank the reviewers for recommendations for final minor revision. We have highlighted in the text where changes have been made.

We look forward to receiving your response in due course.

Best wishes
Debra Bick